Author's response to reviews

Title: The Diagnostic Value of Ultrasonography in Carpal Tunnel Syndrome: A Comparison between Diabetic and Non-Diabetic Patients

Authors:

Nai-Wen Tsai (tsainw@yahoo.com.tw)
Lian-Hui Lee (napaj@adm.cgmh.org.tw)
Chi-Ren Huang (suika68@adm.cgmh.org.tw)
Wen-Neng Chang (cwenneng@ms19.hinet.net)
Hung-Chen Wang (m82whc@yahoo.com.tw)
Yu-Jun Lin (lin-yu-jun@yahoo.com.tw)
Wei-Che Lin (u64lin@yahoo.com.tw)
Tsu-Kung Lin (tklin@adm.cgmh.org.tw)
Ben-Chung Cheng (benzmcl@yahoo.com.tw)
Yu-Jih Su (bensus8@gmail.com)
Chia-Te Kung (kungchiate@gmail.com)
Shu-Fang Chen (fangoe@adm.cgmh.org.tw)
Cheng-Hsien Lu (chiu99@ms44.url.com.tw)

Version: 6 Date: 9 May 2013

Author's response to reviews: see over
Dear Professor Josefino M. Rodis:

Thanks for reviewer’s Leo Visser accept of our response. We have made further revisions for editorial requirement. The revised manuscript entitled as “MS: 1989849118700162 The Diagnostic Value of Ultrasonography in Carpal Tunnel Syndrome: A Comparison between Diabetic and Non-Diabetic Patients”. All of the changes have been underlined with red color to facilitate your review.

Below are our point-by-point responses to the editorial requirements.

**Editorial requirement:**

1. Please move your Author's Contribution section to the correct place in the manuscript. Also, kindly format it according to the instructions for authors and describe exactly what each authors contributed to the manuscript.
   **Answer:** We provided the exactly contribution of each authors and move the Author's Contribution section to the correct place behind the Conclusions section. We also add the List of abbreviations according to the instructions for author’s requirement.

2. Please also ensure that your revised manuscript conforms to the journal style ([http://www.biomedcentral.com/info/ifora/medicine_journals](http://www.biomedcentral.com/info/ifora/medicine_journals)). It is important that your files are correctly formatted.
   **Answer:** We revised our Reference format to the journal style.